

Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2023GL105472

Key Points:

- Increasing OH suggested by Coupled Model Intercomparison Project Phase 6 models can lead to a 5%–7% underestimation in hydrochlorofluorocarbon (HCFC) and hydrofluorocarbon (HFC) emission estimations in 2005
- If there is significant ocean degradation through microbial activity, HCFC, and HFC emissions could be underestimated by up to 10%
- Our study suggests an uncertainty in the combined contribution to global warming from HCFCs and HFCs up to 15%–20% in the 2020s

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

P. Wang, pdwang@mit.edu

Citation:

Wang, P., Solomon, S., Lickley, M., Scott, J. R., Weiss, R. F., & Prinn, R. G. (2023). On the influence of hydroxyl radical changes and ocean sinks on estimated HCFC and HFC emissions and banks. *Geophysical Research Letters*, 50, e2023GL105472. https://doi. org/10.1029/2023GL105472

Received 12 JUL 2023 Accepted 15 SEP 2023

Author Contributions:

Conceptualization: Peidong Wang, Susan Solomon, Megan Lickley Formal analysis: Peidong Wang Funding acquisition: Susan Solomon Methodology: Peidong Wang, Susan Solomon, Megan Lickley, Jeffery R. Scott, Ray F. Weiss Resources: Megan Lickley, Ray F. Weiss, Ronald G. Prinn Software: Peidong Wang, Jeffery R. Scott Supervision: Susan Solomon

© 2023. The Authors. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

On the Influence of Hydroxyl Radical Changes and Ocean Sinks on Estimated HCFC and HFC Emissions and Banks

Peidong Wang¹, Susan Solomon¹, Megan Lickley^{2,3}, Jeffery R. Scott¹, Ray F. Weiss⁴, and Ronald G. Prinn¹

¹Department of Earth, Atmospheric, and Planetary Sciences, Massachusetts Institute of Technology, Cambridge, MA, USA, ²Science, Technology, and International Affairs Program, Georgetown University, Washington, DC, USA, ³Earth Commons, Georgetown University, Washington, DC, USA, ⁴Scripps Institution of Oceanography, University of California San Diego, La Jolla, CA, USA

Abstract Hydrochlorofluorocarbons (HCFCs) and hydrofluorocarbons (HFCs) are potent greenhouse gases regulated under the Montreal Protocol and its amendments. Emission estimates generally use constant atmospheric lifetimes accounting for loss via hydroxyl radical (OH) reactions. However, chemistry-climate models suggest OH increases after 1980, implying underestimated emissions. Further, HCFCs and HFCs are soluble in seawater and could be destroyed through in situ oceanic microbial activity. These ocean sinks are largely overlooked. Using a coupled atmosphere-ocean model, we show that increases in modeled OH imply underestimated HCFC and HFC emissions by ~10% near their respective peak emissions. Our model results also suggest that oceanic processes could lead to up to an additional 10% underestimation in these halocarbon emissions in the 2020s. Ensuring global compliance to the Protocol and accurate knowledge of contributions to global warming from these gases therefore requires understanding of these processes.

Plain Language Summary Man-made hydrochlorofluorocarbons (HCFCs) and hydrofluorocarbons (HFCs) contribute to global warming, prompting worldwide agreement to control the production of these chemicals. It is important to estimate their emissions to ensure global compliance with the agreed phaseout. But correct emission estimation requires knowledge of different loss pathways. One major loss pathway of halocarbons is through chemical reactions with the atmospheric "scrubber" called OH. OH is difficult to measure and usually assumed to be constant with time. But some models suggest OH has increased, which implies increased emissions to match observed abundances. These halocarbons also dissolve into the oceans, where microbes may also metabolize them, but these processes are not included in current emission estimates. We show that if these halocarbons are being consumed in the oceans, this would also lead to an additional underestimation of human emissions. Confidence in the success in the Montreal Protocol and its Kigali amendment to reduce HFCs will therefore require a better understanding of both OH trends and ocean sinks, along with use of HFC and HCFC measurements.

1. Introduction

Chlorofluorocarbons (CFCs) are the main reason for the Antarctic ozone "hole" (e.g., Solomon, 1999). Under the Montreal Protocol on Substances that Deplete the Ozone Layer, the production of these molecules has been strictly regulated since the early 1990s. Hydrochlorofluorocarbons (HCFCs) were used to replace CFCs as they can react with the hydroxyl radical (OH) rapidly in the troposphere, reducing the chlorine reaching the stratosphere, and thus depleting ozone. However HCFC emissions were not globally regulated until the early 2010s, and the amount of chlorine delivered to the stratosphere from HCFCs rose by more than 60% from 1993 to 2020 (Laube & Tegtmeier, 2022). While a complete global phase-out of HCFCs is expected to happen in 2030, their use as feedstocks to manufacture other compounds remains exempted from the Montreal Protocol. In the past two decades, HCFC emissions from feedstocks are believed to have increased by five-fold (Daniel & Reimann, 2022) and leakage rates are higher than expected (Western et al., 2022), prompting discussions about narrowing feedstock exemptions (Andersen et al., 2021). HCFCs were replaced by hydrofluorocarbons (HFCs). HFCs are not ozone depletors but are potent greenhouse gases that have global warming potentials ranging from thousands to tens of thousands times that of carbon dioxide on a mass basis (Liang & Rigby, 2022). The Kigali Amendment to the Montreal Protocol to regulate HFCs entered into force in 2019. The amendment is expected to lead to a global emissions peak in the 2030s. If effective mitigation of HFCs occurs worldwide, the projected global





Visualization: Peidong Wang Writing – original draft: Peidong Wang Writing – review & editing: Peidong Wang, Susan Solomon, Megan Lickley, Jeffery R. Scott, Ray F. Weiss, Ronald G. Prinn warming, compared to continuing production of HFCs, will be reduced by up to 0.4°C by the end of 2100 (Velders et al., 2022).

The primary loss pathway for HCFCs and HFCs is through reaction with OH. But due to a lack of direct measurements of global-scale OH concentrations and uncertainties in the inferred OH concentration based on methyl chloroform observations (Prinn et al., 2018), current emission estimations for HCFCs and HFCs generally assume OH is seasonally-varying but annually-repeating over time. Here we consider time-varying potential trends in OH. Turner et al. (2017) found that changes in methane may be explained by an increase in OH from 1980 to 2005 and a decrease afterward. A similar decreasing trend in OH after 2005 was also found in several inversion studies (Montzka et al., 2011; Rigby et al., 2017). In contrast, while a suite of Coupled Model Intercomparison Project Phase 6 models suggest a 10% increase in OH from 1980 to 2005 similar to Turner et al. (2017), they indicate ongoing increases afterward (Stevenson et al., 2020). Recently, using methyl chloroform with 3D model inversions incorporating observed meteorology, Patra et al. (2021) also deduced a significant OH increase after 2005. We explore increasing OH trends from 1980 to 2005 and increasing or decreasing trends thereafter.

Halocarbons (mainly CFC-11 and CFC-12) have a long history of use as inert tracers to study ocean circulation (e.g., Bullister & Weiss, 1983; Ito et al., 2004; Romanou et al., 2017; Waugh, 2014). In an earlier study (Wang et al., 2021), we showed that ocean uptake can affect the inferred atmospheric residence times of CFC-11 and CFC-12, therefore affecting their emission estimates. Similarly, the ocean represents a potential loss pathway for HCFCs and HFCs since they are also soluble in seawater. However, the ocean could be a larger sink for these gases than previously thought, because certain microbes can feed on hydrogen-bearing compounds (Dworkin et al., 2006), such as methyl halides (Cox et al., 2012; Tokarczyk, Goodwin, & Saltzman et al., 2003; Tokarczyk, Saltzman, et al., 2003). Similarly, aerobic oxidation can also degrade certain HCFCs and HFCs (Chang & Criddle, 1995; DeFlaun et al., 1992; Streger et al., 1999). But due to limited measurements, ocean degradation of HCFCs and HFCs via biological processes has not been considered in previous estimations of ocean partial lifetimes (Liang & Rigby, 2022; Yvon-Lewis & Butler, 2002). Hydrolysis is another ocean degradation process for halogenated species such as methyl chloroform (Wennberg, 2004), but is considered minor for HCFCs and HFCs (Yvon-Lewis & Butler, 2002). Here we explicitly evaluate the atmospheric residence time, treating the ocean destruction and reservoir both as sinks insofar as the atmosphere (as appropriate for deriving emissions) is concerned; previous work has often used a different definition of ocean destruction only, which leads to much smaller impacts (Yvon-Lewis & Butler, 2002).

Given debate regarding feedstock emissions for HCFCs and new regulations on HFCs, it is crucial to consider uncertainties in emission estimates inferred from observed atmospheric abundance data to ensure the continued success of the Montreal Protocol and its amendments. In this study, we quantify uncertainties in emission estimations and banks due to different OH scenarios (time-varying OH trends vs. fixed OH) and ocean scenarios (only involving physical dissolution or with additional microbial ocean degradation) for three HCFCs (HCFC-141b, HCFC-142b, and HCFC-22) and three HFCs (HFC-125, HFC-134a, and HFC-23) whose seawater solubilities are provided by Li et al. (2019).

2. Data and Methods

We couple the ocean representation of the Massachusetts Institute of Technology General Circulation Model (MITgcm, Marshall, Adcroft, et al., 1997; Marshall, Hill, et al., 1997) with the atmosphere from the Advanced Global Atmospheric Gases Experiment (AGAGE) 12-box model (Rigby et al., 2013). This coupled model runs independently from the start of emissions for each species to 2050.

2.1. Ocean Representation

We use a similar MITgcm ocean configuration as Wang et al. (2021), adopting a $2.8^{\circ} \times 2.8^{\circ}$ horizontal resolution and 15 vertical layers down to 5 km depth. Despite its coarse resolution, this configuration simulates the oceanic distributions and the overall ocean burdens of CFCs reasonably well (Wang et al., 2021). The ocean is forced with monthly climatologies of wind stress and buoyancy fluxes. A detailed description of each of the forcing fields is discussed in Wang et al. (2021). Prior to the introduction of halocarbon emissions, the ocean was spun-up for 5,900 years with these climatological forcings to effectively reach a steady state. We only consider climatological forcing in this study since our earlier study showed that future climate change had a rather small impact on the global CFC ocean burden (although it does affect the timing for the ocean to switch from a net sink to a net source (Wang et al., 2021)).

Li et al. (2019) report the Ostwald solubilities as functions of temperature and salinity for several HCFCs and HFCs. Freshwater solubilities are compared with other studies using various methods, and good agreement is achieved for most of the species except HFC-125 (Li et al., 2019). These temperature- and salinity-dependent solubilities are then fitted to a polynomial that has the format used by the MITgcm. Table S1 in Supporting Information S1 shows the fitted coefficients for HCFCs and HFCs used in this study. Figure S1 in Supporting Information S1 plots the solubility for each species as a function of temperature and salinity. These molecules are generally more soluble in colder and less salty conditions.

The Schmidt number (Sc) is used in the MITgcm to calculate the gas transfer velocity. It characterizes the ratio between the kinematic viscosity of the solution and the diffusion coefficient of the gas in the solution (Wanninkhof, 1992). We follow a similar process to obtain seawater Sc values for HCFCs and HFCs as in Wanninkhof (1992). Sc in the MITgcm is parameterized only as a function of temperature at a constant salinity of 35% (per mille); see Text S1, Table S2, and Figure S2 in Supporting Information S1 for details.

Several studies indicate that aerobic microbial processes in the ocean can destroy certain HCFCs and HFCs (Chang & Criddle, 1995; DeFlaun et al., 1992; Streger et al., 1999). However, ocean degradation rates can vary significantly depending upon spatial and temporal variations in biology and other properties such as oxidation state. Due to limited knowledge of the ocean degradation processes, we perform several ocean scenarios to explore various assumptions: (a) no ocean uptake (noOCN), (b) with ocean dissolution but without any degradation (OCN), and (c) with an e-folding ocean degradation timescale at 1 year (degradeOCN). We choose 1 year as the upper limit of ocean degradation losses because HCFC-22, which experiences faster decay than other species (Chang & Criddle, 1995), degrades with a timescale on the order of a year in the ocean mixed layer (detailed calculation in Text S2 in Supporting Information S1). We also present results for ocean degradation with a timescale of 5 years to illustrate the differences found between no biological degradation and a 1 year degradation timescale.

2.2. Atmosphere Representation

We use the AGAGE 12-box model as the atmospheric portion of our coupled ocean-atmosphere framework. This model has three vertical layers (surface to 500, 500–200, and above 200 hPa) and four zonally-averaged latitudinal bands (90°–30°N, 30°N–equator, equator–30°S, and 30°–90°S). It employs monthly OH concentrations and transport parameters for each box. A detailed description is provided in Rigby et al. (2013). Note that we do not consider year-to-year changes in the atmospheric transport parameters, but such interannual dynamical changes can also influence variations in inferred lifetimes and emissions (Ko et al., 2013).

Historical input emissions for HCFCs and HFCs between the 1970s and 2021 are from the posterior mean emission (later referred to as posterior emission) obtained by inverse modeling of the AGAGE measured mole fractions of each species combined with a 12-box model that assumes fixed OH and constant lifetimes from the World Meteorological Organization 2022 report (Laube & Tegtmeier, 2022; Liang & Rigby, 2022; M. Rigby et al., 2013). However, anthropogenic emissions for HCFC-22 and HFC-23 started long before the AGAGE observations began and posterior emissions are available. For these two species, we run the coupled ocean-atmosphere model starting in the 1940s with zero initial concentrations in both the ocean and atmosphere, employing a priori emissions (Miller et al., 2010) to produce reasonable tracer distributions, and then switch to the posterior emission suffer the 1970s. Here we focus on the period when we switch to the posterior emission inputs. Figure S3 in Supporting Information S1 shows the model simulated atmospheric mole fractions for all the species used in this analysis using the posterior emissions compared to the AGAGE observations (Prinn et al. (2018) updated through 2021 using data from https://agage.mit.edu/data), and they show good agreement.

Future emissions for HFC-125 and HFC-134a are from the upper limit of the Kigali Amendment projection from Velders et al. (2022). Future emissions for HCFC-141b, HCFC-142b, and HCFC-22 are obtained from the inverse study of the projected mole fractions from Daniel and Reimann (2022) using a one-box model assuming a perfectly mixed atmosphere and constant lifetimes. HFC-23 is mainly a by-product of HCFC-22 production, and we assume future HFC-23 emissions follow the same rate of change as HCFC-22 (neglecting potential differences during leakage from banks such as refrigeration). To prevent discontinuity in the input emissions, we scale

global projected emissions from 2022–2050 to emissions in 2021, and apply this scaling to generate a smooth transition from the historical to future emissions.

Different OH scenarios are considered in order to infer HCFC and HFC atmospheric residence times and emissions. In the "fixOH" scenario, we use repeating monthly OH concentrations every year from Rigby et al. (2013). In the "varyOH" scenario, OH does not change before 1980, but it increases by 10% between 1980 and 2005 (in each box and month). Future projections of OH to 2050 vary greatly among different models (Murray et al., 2021). We therefore apply a $\pm 15\%$ change of OH between 2005 and 2050 (spanning the range from different warming scenarios (Murray et al., 2021)). Figure S4 in Supporting Information S1 summarizes different OH assumptions used in this study. Note that the absolute amount of OH concentration from inversions based on methyl chloroform can have large uncertainties (Rigby et al., 2017). These absolute uncertainties largely cancel insofar as they exist in both the "fixOH" and "varyOH" scenarios, and only the relative change in the emissions between different OH scenarios is affected.

2.3. Inferring the Atmospheric Residence Times and the Emissions

We refer to the e-folding decay timescale for each species as the atmospheric residence time in this study since the residence time involves not only atmospheric loss rates (i.e., reaction with OH) but also ocean processes (such as ocean uptake and outgassing) characterized by:

$$\tau(t) = \frac{B(t)}{L_{\text{OH}}(t) + L_{\text{OCN}}(t) + L_{\text{OTHER}}(t)}$$
(1)

where *B* is the total atmospheric burden, L_{OH} , L_{OCN} , and L_{OTHER} are the total loss rates due to OH, ocean sinks, and other photochemistry (largely stratospheric), respectively. L_{OCN} can be negative when the ocean becomes supersaturated and returns gas back to the atmosphere.

To infer global emissions, we use different residence times τ calculated from various combinations of ocean and OH scenarios, using the equation:

$$E(t) = \frac{dB(t)}{dt} + \frac{B(t)}{\tau(t)}$$
(2)

where the atmospheric burden B is the simulated "noOCN" and "fixOH" case, that is, what we would have "observed" in the atmosphere in the absence of time-varying OH and ocean uptake.

3. Results

3.1. Relative Magnitudes of Different Loss Pathways

We first quantify a baseline magnitude (without varying OH and neglecting ocean degradation) of each loss pathway for HCFCs and HFCs (Figure S5 in Supporting Information S1). Without considering ocean degradation, the global ocean sink and stratospheric loss through photochemistry are generally about an order of magnitude smaller than loss due to OH. However, HFC-23 has a slow reaction with OH and an atmospheric lifetime greater than 200 years (while other species considered here display decadal lifetimes), and ocean loss is comparable to OH loss for this molecule. For all species, loss rates due to ocean sinks differ between the two hemispheres. Ocean uptake in the Southern Hemisphere is twice that in the Northern Hemisphere (Figures S6 and S7 in Supporting Information S1), due in part to the larger open ocean area and enhanced upwelling of undersaturated waters driven by strong surface wind stress. Therefore, loss to the ocean in the 30°–90°S latitude band accounts for 10%–20% of the loss to OH in that region for most of the species considered here. Globally summed, loss to the ocean is less than about 5% of the total loss to OH for HCFCs and HFCs in this analysis, with the important exception of HFC-23.

3.2. Inferred Atmospheric Residence Times

Figure 1 shows the inferred atmospheric residence times for each species. Species other than HCFC-22 and HFC-23 have close to zero emissions in the 1980s, reflected in steep changes in the residence times in the first



Geophysical Research Letters



Figure 1. Global atmospheric residence times for all the species used in this study. Gray dotted lines are the steady state lifetimes calculated from the AGAGE 12-box model based on atmospheric removal rates from the WMO (2022) report. Black dashed lines are inferred from a model simulation with fixed OH over time and without ocean uptake. Blue solid lines are inferred from models with fixed OH over time but including ocean dissolution without degradation, while blue dashed lines assume a 5 years ocean degradation timescale, blue shadings denote the case for ocean degradation timescale up to 1 year. Red solid lines indicate a 10% increase in OH from 1980 to 2005 and are constant beyond 2005, and the red shadings indicate $\pm 15\%$ OH changes from 2005 to 2050. The purple dash-dotted lines combine the increasing OH (both 10% increase in 1980–2005 and 15% increase in 2005–2050) with a 1 year ocean degradation timescale to define a lower limit range.

few years. The impact of the ocean on the inferred residence times is largest in the early period, when the ocean is less saturated and has a greater ability to dissolve these gases. Overall, the ocean shortens the inferred residence times during most of the time before 2050, as the ocean is still a net sink for these species globally (although earlier ocean outgassing is occurring regionally in the tropical and coastal upwelling zones). However, an exception is HCFC-22, for which the ocean starts to become a net source of emission in the 2040s globally (Figure S5h in Supporting Information S1), and hence its residence times increase afterward. Without ocean degradation, adding the ocean shortens the inferred residence times by up to 2% before 2020 for most of the species, and 15% for HFC-23. As the ocean becomes more saturated over time, the difference in the inferred residence times with versus without the ocean shrinks.

However, if a dissolved HCFC or HFC is degraded in the ocean, the ocean will have a greater ability to uptake these gases and will be difficult to oversaturate relative to its declining trend in the atmosphere. By 2020, assuming halocarbons degrade in the ocean with a timescale of 5 years, their residence times are shortened by 2%-5% compared to the noOCN case for most species (and a 25% decrease in residence time for HFC-23). And if the ocean degradation timescale is 1 year, we obtain a 3%-10% decrease in the inferred residence times for most of the species (and a 40% decrease for HFC-23).

Increasing OH concentrations would also shorten residence times in the atmosphere. A 10% increase in the OH concentration from 1980 to 2005 leads to about an 8% decrease in the inferred residence time by 2005 for all the species in this study. After 2005, if OH increased by another 15% by 2050 as some models indicate, this would result in a 15%–20% decrease in the inferred residence time by the end of 2050 compared to the fixed OH scenario. However, if OH decreases after 2005, the inferred residence time would return to the fixed OH scenario by the end of 2040. The purple dot-dashed lines in Figure 1 show the lower limits of residence times obtained by combining both a 1 year ocean degradation timescale and an increasing OH of 25% from 1980 to 2050.



Geophysical Research Letters

10.1029/2023GL105472



Figure 2. Inferred global emissions for all the species modeled in this study, based on the different residence times for the cases considered in Figure 1.

3.3. Inferred Global Emissions and Bank Estimations

Figure 2 shows the inferred global emissions for HCFCs and HFCs using different inferred atmospheric residence times obtained from Section 3.2 (changes in the emissions shown in Figure S8 in Supporting Information S1). Comparing cases with ocean uptake (no ocean degradation) to those without ocean uptake implies that neglect of the ocean leads to a 1%–2% underestimation in the inferred emissions for all three HCFCs and HFC-23, and less than 1% for HFC-125 and HFC-134a around 2020. If there is a 5 years ocean degradation timescale, we obtain another 2%–3% underestimation in the inferred emissions for HCFCs and HFC-23 is obtained, with less than 5% underestimation for HFC-125 and HFC-134a. Ocean uptake can significantly decrease the HFC-23 atmospheric residence time, but it is still above 100 years, so the relative impact on its inferred emission is not as large as the impact on its residence time.

With a 10% increase in OH concentration only from 1980 to 2005, we obtain a 5%-7% underestimation in the inferred global emissions by the end of 2005 for all three HCFCs compared to the case of the fixed OH scenario. For HFCs, there is a 1%-5% underestimation in the inferred emissions by 2005. There is a large variation in the inferred emissions after 2005 given different OH scenarios, and the corresponding variations in the emission due to different OH scenarios by the end of 2050 can be up to 40%. Considering both a 1 year ocean degradation timescale and increasing OH would imply an underestimate of the emissions, by around 10%-20% during the peak of emissions for HCFCs and HFCs.

By weighting the emissions of each species by its global warming potential compared to CO_2 (on a 100-year horizon; Burkholder & Hodnebrog, 2022), we calculate the CO_2 -equivalent emissions summed from all six HCFCs and HFCs in this study (Figure 3; Figure S9 in Supporting Information S1 breaks the summed emission into different species). During the peak total emission period in the 2020s, this would underestimate the emission by 1% if one does not include the ocean uptake (and by 6% if the ocean degrades at 1 year). And if OH keeps increasing from 1980 to the 2020s, using fixed OH would underestimate the emission by another 7%. And combining ocean degradation and increase in OH could underestimate the CO_2 -equivalent emission by up to 15%–20% in the 2020s.



19448007, 2023, 18, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2023GL105472, Wiley Online Library on [04/06/2025]. See the Terms



Figure 3. Summed emissions of all the hydrochlorofluorocarbons and hydrofluorocarbons in this study weighted by their global warming potentials. The absolute emissions are shown in panel (a), and the differences between the emissions inferred from transient residence times and steady state lifetimes (assuming no ocean and fixed OH) are shown in panel (b).

The estimated bank using a top-down method is the cumulative difference between the reported production and the inferred emission (calculated from Equation 2). If we assume that production has been accurately reported, then with higher inferred emissions resulting in banks depleting faster than currently expected. We quantify the resulting impact on banks by integrating the inferred emissions (from Figure 2) over time. Figure S10 in Supporting Information S1 shows the difference between HCFC and HFC banks estimates calculated from the (noOCN and fixOH) steady state residence times based on atmospheric removal rates reported in Burkholder and Hodnebrog (2022) and banks calculated from transient residence times. In the current decade, if we do not consider the ocean processes, we would overestimate the bank by 22-101 Gg for HCFC-141b, 8-40 Gg for HCFC-142b, 197–1,090 Gg for HCFC-22, 3–9 Gg for HFC-125, 40–152 Gg for HFC-134a, and 6–34 Gg for HFC-23 (the range spans no ocean degradation to a 1 year decay timescale). If we do not consider the increase in OH from 1980 to 2005, we would have overestimated the banks by 76-140 Gg for HCFC-141b, 35-63 Gg for HCFC-142b, 537-940 Gg for HCFC-22, 16-45 Gg for HFC-125, 148-340 Gg for HFC-134a, and 2-4 Gg for HFC-23 in the 2020s (the range spans continued OH increases or decreasing after 2005). Ocean uptake/loss and increasing OH would add to each other, resulting in an even larger overestimation in the global banks for these molecules (shown as the purple dash-dotted line in Figure S10 in Supporting Information S1). Alternatively, if we assume that bank estimates are accurate, higher emissions could be an indication of higher than reported production, where emissions in the year of production could be anywhere between 6% and 83%, depending on the application and gas (Lickley et al., 2022), with the remaining 17%–94% of production going into the banks. This would imply that banks are in fact higher than published estimates suggest, and the bias could be an order of magnitude (if initial releases are closer to 6%) and of the opposite sign than that shown in Figure S10 in Supporting Information S1. This wide range of uncertainty underscores the need to account for the ocean and OH lifetime effects in future bank assessments.

4. Conclusions and Discussions

In this study, we have explored the uncertainties from the ocean uptake and OH sinks for several Montreal Protocol-regulated molecules. We show that if one does not consider oceanic dissolution, and assumes no oceanic degradation, this would underestimate the emissions of HCFCs and HFCs by about 1%–2% in the current decade. This impact is rather small. Due to limited knowledge of ocean degradation rates, these species are often treated as conservative tracers within the ocean (Liang & Rigby, 2022; Yvon-Lewis & Butler, 2002). But evidence suggests certain species can be destroyed in the ocean (Chang & Criddle, 1995; DeFlaun et al., 1992; Streger et al., 1999). If there is a significant ocean degradation pathway, the emission of HCFCs and HFCs would be underestimated by up to 10% depending on how fast these species are destroyed in the ocean. Climate change can add more uncertainty in emission estimations by directly affecting HCFC and HFC solubilities, ocean circulation (Wang et al., 2021) and/or influencing the oceanic biological activity (Kirchman et al., 2009).

WANG ET AL.

Wiley Online Library for rules of use; OA

d by the :

applicable Creative

nons Licens

19448007, 2023, 18, Downloaded from https://agupubs.online

.wiley

.com/doi/10.1029/2023GL105472, Wiley Online Library on [04/06/2025]. See the Terms and Conditions

(https://onli

library.wiley.com/tem

ditions) on Wiley Online Library

for rules of use; OA

articles are governed by the applicable Creative Com

The main loss pathways for HCFCs and HFCs are through their reactions with OH. Models and inversions consistently suggest a 10% increase in OH from 1980 to 2005 (Stevenson et al., 2020; Turner et al., 2017), but current emission estimations for HCFCs and HFCs are based on fixed OH concentrations over time. If one assumes that OH is fixed, whereas it is actually increasing, emissions would be underestimated by 5%-7% by 2005. Future emissions estimation also depends strongly on an accurate estimation of OH concentration changes out to 2050.

The relative importance of including the ocean and OH trend for emission estimation is determined by how these processes compete. The ocean plays a significant role for HFC-23, whose reaction rate with OH is more than 20 times slower than the other modeled species, even though it is 2-5 times less soluble in seawater. For other species whose reaction rates with OH are faster: the ocean plays a more significant role for HCFC-141b and HCFC-22 due to their relatively high solubilities compared to HCFC-142b, HFC-125, and HFC-134a.

Although new production of HCFCs is now controlled, utilizing these species as feedstocks is exempted from the Montreal Protocol, based on the assumption that feedstock emissions should be small. By 2019, however, up to 8% and 33% of the global emissions of HCFC-22 and HCFC-142b appear to be from the feedstock emissions (Daniel & Reimann, 2022). The feedstock market has increased sharply in the past decade and is likely to continue increasing if there are no further actions on the feedstock exemption (Andersen et al., 2021). Inferences of banks and emissions (including feedstock emissions) depend on knowledge of residence times, for which OH trends and possible ocean degradation are important uncertainties. HCFC and HFC total emissions represent about 1.8 Gt CO₂-equivalent per year, or about 5% of total emissions of all global warming agents (IEA, 2022). Our study suggests a total uncertainty in their contribution to warming of up to 15%-20% in the current decade and draws attention to the need for additional information on oceanic degradation processes and on OH trends to better quantify the atmospheric emissions of these compounds. This will add a perspective that has been overlooked in the past to the discussion of ensuring the success of the Montreal Protocol and its following amendments.

Data Availability Statement

The coupled MITgcm-12 box model code and output, code to generate figures, and tabulated files containing the atmospheric residence times for each species under different OH and ocean scenarios are available at https://doi. org/10.5281/zenodo.8307171 (Wang et al., 2023).

References

- Andersen, S. O., Gao, S., Carvalho, S., Ferris, T., Gonzalez, M., Sherman, N. J., et al. (2021). Narrowing feedstock exemptions under the Montreal Protocol has multiple environmental benefits. Proceedings of the National Academy of Sciences of the United States of America, 118(49), e2022668118. https://doi.org/10.1073/pnas.2022668118
- Bullister, J. L., & Weiss, R. F. (1983). Anthropogenic chlorofluoromethanes in the Greenland and Norwegian seas. Science, 221(4607), 265–268. https://doi.org/10.1126/science.221.4607.265
- Burkholder, J. B., & Hodnebrog, Ø. (2022). Scientific assessment of ozone depletion: 2022, annex: Summary of abundances, lifetimes, ODPs, REs, GWPs, and GTPs (No. (278), p. 509). World Meteorological Organization.
- Chang, W., & Criddle, C. S. (1995). Biotransformation of HCFC-22, HCFC-142b, HCFC-123, and HFC-134a by methanotrophic mixed culture MM1. Biodegradation, 6(1), 1-9. https://doi.org/10.1007/BF00702293
- Cox, M. J., Schäfer, H., Nightingale, P. D., McDonald, I. R., & Murrell, J. C. (2012). Diversity of methyl halide-degrading microorganisms in oceanic and coastal waters. FEMS Microbiology Letters, 334(2), 111-118. https://doi.org/10.1111/j.1574-6968.2012.02624.x
- Daniel, J., & Reimann, S. (2022). Scientific assessment of ozone depletion: 2022. In Chapter 7: Scenarios and information for policymakers (No. (278), p. 509). World Meteorological Organization.
- DeFlaun, M. F., Ensley, B. D., & Steffan, R. J. (1992). Biological oxidation of hydrochlorofluorocarbons (HCFCs) by a methanotrophic bacterium. Nature Biotechnology, 10(12), 1576-1578. https://doi.org/10.1038/nbt1292-1576
- Dworkin, M., Falkow, S., Rosenberg, E., Schleifer, K.-H., & Stackebrandt, E. (Eds.) (2006). The prokaryotes. Springer New York. https://doi. org/10.1007/0-387-30745-1
- IEA. (2022). Global energy review: CO2 emissions in 2021. International Energy Agency. Retrieved from https://www.iea.org/reports/globalenergy-review-co2-emissions-in-2021-2
- Ito, T., Marshall, J., & Follows, M. (2004). What controls the uptake of transient tracers in the Southern Ocean? Global Biogeochemical Cycles, 18(2), GB2021. https://doi.org/10.1029/2003GB002103
- Kirchman, D. L., Morán, X. A. G., & Ducklow, H. (2009). Microbial growth in the polar oceans Role of temperature and potential impact of climate change. Nature Reviews Microbiology, 7(6), 451-459. https://doi.org/10.1038/nrmicro2115
- Ko, M., Newman, P., Reimann, S., & Strahan, S. (2013). SPARC report on the lifetimes of stratospheric ozone-depleting Substances, their replacements, and related species. WCRP.
- Laube, J., & Tegtmeier, S. (2022). Scientific assessment of ozone depletion: 2022. In Chapter 1: Update on ozone-depleting substances (ODSs) and other gases of interest to the Montreal Protocol (No. (278), p. 509). World Meteorological Organization.
- Li, P., Mühle, J., Montzka, S. A., Oram, D. E., Miller, B. R., Weiss, R. F., et al. (2019). Atmospheric histories, growth rates and solubilities in seawater and other natural waters of the potential transient tracers HCFC-22, HCFC-141b, HCFC-142b, HFC-134a, HFC-125, HFC-23, PFC-14 and PFC-116. Ocean Science, 15(1), 33-60. https://doi.org/10.5194/os-15-33-2019

P.W., M. L., J. S., and S. S. gratefully acknowledge Grant 2128617 from the Atmospheric Chemistry Division of the National Science Foundation and support of VoLo foundation. AGAGE (R.G.P., R.F.W.) is supported principally by multi-year NASA Grants to MIT and SIO. Support also comes from BEIS (UK), NOAA, and CSIRO (Australia). The authors appreciate the emissions data from Matthew Rigby (University of Bristol), John Daniel (NOAA), and Guus Velders (RIVM and Utrecht University). The authors also appreciate helpful comments on an earlier draft by Stephen Montzka (NOAA).

- Liang, Q., & Rigby, M. (2022). Scientific assessment of ozone depletion: 2022. In *Chapter 2: Hydrofluorocarbons* (No. (278), p. 509). World Meteorological Organization.
- Lickley, M. J., Daniel, J. S., Fleming, E. L., Reimann, S., & Solomon, S. (2022). Bayesian assessment of chlorofluorocarbon (CFC), hydrochlorofluorocarbon (HCFC) and Halon banks suggest large reservoirs still present in old equipment. *Atmospheric Chemistry and Physics*, 22(17), 11125–11136. https://doi.org/10.5194/acp-22-11125-2022
- Marshall, J., Adcroft, A., Hill, C., Perelman, L., & Heisey, C. (1997). A finite-volume, incompressible Navier Stokes model for studies of the ocean on parallel computers. *Journal of Geophysical Research*, 102(C3), 5753–5766. https://doi.org/10.1029/96JC02775
- Marshall, J., Hill, C., Perelman, L., & Adcroft, A. (1997). Hydrostatic, quasi-hydrostatic, and nonhydrostatic ocean modeling. *Journal of Geophysical Research*, 102(C3), 5733–5752. https://doi.org/10.1029/96JC02776
- Miller, B. R., Rigby, M., Kuijpers, L. J. M., Krummel, P. B., Steele, L. P., Leist, M., et al. (2010). HFC-23 (CHF3) emission trend response to HCFC-22 (CHCIF2) production and recent HFC-23 emission abatement measures. *Atmospheric Chemistry and Physics*, 10(16), 7875–7890. https://doi.org/10.5194/acp-10-7875-2010
- Montzka, S. A., Krol, M., Dlugokencky, E., Hall, B., Jöckel, P., & Lelieveld, J. (2011). Small interannual variability of global atmospheric hydroxyl. Science, 331(6013), 67–69. https://doi.org/10.1126/science.1197640
- Murray, L. T., Fiore, A. M., Shindell, D. T., Naik, V., & Horowitz, L. W. (2021). Large uncertainties in global hydroxyl projections tied to fate of reactive nitrogen and carbon. *Proceedings of the National Academy of Sciences of the United States of America*, 118(43), e2115204118. https://doi.org/10.1073/pnas.2115204118
- Patra, P. K., Krol, M. C., Prinn, R. G., Takigawa, M., Mühle, J., Montzka, S. A., et al. (2021). Methyl chloroform continues to constrain the hydroxyl (OH) variability in the troposphere. *Journal of Geophysical Research: Atmospheres*, 126(4), e2020JD033862. https://doi. org/10.1029/2020JD033862
- Prinn, R. G., Weiss, R. F., Arduini, J., Arnold, T., DeWitt, H. L., Fraser, P. J., et al. (2018). History of chemically and radiatively important atmospheric gases from the advanced global atmospheric gases experiment (AGAGE). *Earth System Science Data*, 10(2), 985–1018. https:// doi.org/10.5194/essd-10-985-2018
- Rigby, M., Montzka, S. A., Prinn, R. G., White, J. W. C., Young, D., O'Doherty, S., et al. (2017). Role of atmospheric oxidation in recent methane growth. Proceedings of the National Academy of Sciences of the United States of America, 114(21), 5373–5377. https://doi.org/10.1073/ pnas.1616426114
- Rigby, M., Prinn, R. G., O'Doherty, S., Montzka, S. A., McCulloch, A., Harth, C. M., et al. (2013). Re-evaluation of the lifetimes of the major CFCs and CH3CCl3 using atmospheric trends. Atmospheric Chemistry and Physics, 13(5), 2691–2702. https://doi.org/10.5194/acp-13-2691-2013
- Romanou, A., Marshall, J., Kelley, M., & Scott, J. (2017). Role of the ocean's AMOC in setting the uptake efficiency of transient tracers: AMOC and tracer uptake. *Geophysical Research Letters*, 44(11), 5590–5598. https://doi.org/10.1002/2017GL072972
- Solomon, S. (1999). Stratospheric ozone depletion: A review of concepts and history. *Reviews of Geophysics*, 37(3), 275–316. https://doi.org/10.1029/1999RG900008
- Stevenson, D. S., Zhao, A., Naik, V., O'Connor, F. M., Tilmes, S., Zeng, G., et al. (2020). Trends in global tropospheric hydroxyl radical and methane lifetime since 1850 from AerChemMIP. Atmospheric Chemistry and Physics, 20(21), 12905–12920. https://doi.org/10.5194/ acp-20-12905-2020
- Streger, S. H., Condee, C. W., Togna, A. P., & DeFlaun, M. F. (1999). Degradation of hydrohalocarbons and brominated compounds by methaneand propane-oxidizing bacteria. *Environmental Science & Technology*, 33(24), 4477–4482. https://doi.org/10.1021/es9907459
- Tokarczyk, R., Goodwin, K. D., & Saltzman, E. S. (2003). Methyl chloride and methyl bromide degradation in the Southern Ocean. *Geophysical Research Letters*, 30(15), 1808. https://doi.org/10.1029/2003GL017459
- Tokarczyk, R., Saltzman, E. S., Moore, R. M., & Yvon-Lewis, S. A. (2003). Biological degradation of methyl chloride in coastal seawater. Global Biogeochemical Cycles, 17(2), 1057. https://doi.org/10.1029/2002GB001949
- Turner, A. J., Frankenberg, C., Wennberg, P. O., & Jacob, D. J. (2017). Ambiguity in the causes for decadal trends in atmospheric methane and hydroxyl. Proceedings of the National Academy of Sciences of the United States of America, 114(21), 5367–5372. https://doi.org/10.1073/ pnas.1616020114
- Velders, G. J. M., Daniel, J. S., Montzka, S. A., Vimont, I., Rigby, M., Krummel, P. B., et al. (2022). Projections of hydrofluorocarbons (HFC) emissions and the resulting global warming based on recent trends in observed abundances and current policies. *Atmospheric Chemistry and Physics*, 22(9), 6087–6101. https://doi.org/10.5194/acp-22-6087-2022
- Wang, P., Scott, J. R., Solomon, S., Marshall, J., Babbin, A. R., Lickley, M., et al. (2021). On the effects of the ocean on atmospheric CFC-11 lifetimes and emissions. *Proceedings of the National Academy of Sciences of the United States of America*, 118(12), e2021528118. https://doi. org/10.1073/pnas.2021528118
- Wang, P., Solomon, S., Lickley, M., Scott, J. R., Weiss, R. F., & Prinn, R. G. (2023). Model and code for "On the influence of hydroxyl radical and ocean sinks on estimated HCFC and HFC emissions and banks" (Version 2) [Dataset]. Zenodo. https://doi.org/10.5281/zenodo.8307171
- Wanninkhof, R. (1992). Relationship between wind speed and gas exchange over the ocean. *Journal of Geophysical Research*, 97(C5), 7373–7382. https://doi.org/10.1029/92JC00188
- Waugh, D. W. (2014). Changes in the ventilation of the southern oceans. Philosophical Transactions of the Royal Society A: Mathematical, Physical & Engineering Sciences, 372(2019), 20130269. https://doi.org/10.1098/rsta.2013.0269
- Wennberg, P. O. (2004). Recent changes in the air-sea gas exchange of methyl chloroform. *Geophysical Research Letters*, 31(16), L16112. https://doi.org/10.1029/2004GL020476
- Western, L. M., Redington, A. L., Manning, A. J., Trudinger, C. M., Hu, L., Henne, S., et al. (2022). A renewed rise in global HCFC-141b emissions between 2017–2021. Atmospheric Chemistry and Physics, 22(14), 9601–9616. https://doi.org/10.5194/acp-22-9601-2022
- Yvon-Lewis, S. A., & Butler, J. (2002). Effect of oceanic uptake on atmospheric lifetimes of selected trace gases. Journal of Geophysical Research, 107(D20), 4414. https://doi.org/10.1029/2001JD001267

References From the Supporting Information

Belessiotis, V., Kalogirou, S., & Delyannis, E. (2016). Water, the raw material for desalination. Elsevier.

- Bigg, P. H. (1967). Density of water in SI units over the range 0-40°C. British Journal of Applied Physics, 18(4), 521–525. https://doi. org/10.1088/0508-3443/18/4/315
- England, M. H., Garçon, V., & Minster, J.-F. (1994). Chlorofluorocarbon uptake in a world ocean model: 1. Sensitivity to the surface gas forcing. Journal of Geophysical Research, 99(C12), 25215–25233. https://doi.org/10.1029/94JC02205

- Li, P., & Tanhua, T. (2021). Medusa–aqua system: Simultaneous measurement and evaluation of novel potential halogenated transient tracers HCFCs, HFCs, and PFCs in the ocean. *Ocean Science*, *17*(2), 509–525. https://doi.org/10.5194/os-17-509-2021
- Millero, F. J., & Poisson, A. (1981). International one-atmosphere equation of state of seawater. Deep-Sea Research, Part A: Oceanographic Research Papers, 28(6), 625–629. https://doi.org/10.1016/0198-0149(81)90122-9
- Reid, R. C., Prausnitz, J. M., & Poling, B. E. (1987). The properties of gases and liquids. McGraw Hill.
- Wilke, C. R., & Chang, P. (1955). Correlation of diffusion coefficients in dilute solutions. AIChE Journal, 1(2), 264–270. https://doi.org/10.1002/ aic.69001022